Manuscript Title

Inferior wall Myocardial Infarction Masquerading like Peptic Ulcer Disease

Author(s)

Ibrahim Mohsin¹, Zain Fatiwala², Habeeb Ahmed³

¹Kaaj Health Care, USA

²Heart and Vascular Institute, USA

³Regional Medical Center, USA

Summary

The manuscript was received on December 05, 2018 and was peer reviewed by one reviewer and an editor.

The initial recommendation of Revise and Resubmit was made on February 14, 2019. The first revision was submitted on February 28, 2019 and was re-evaluated by the editor. The manuscript was accepted for publication on April 08, 2019.

Peer Review Comments

Peer Reviewer 1:

The manuscript is of significance to readers but this case is not a rare presentation. I have also two concerns about it.

- 1. About the diagnosis, the inferior myocardial infarction is conjectural due to absence of markers of myocardial necrosis.
- 2.In clinical practice, ECG is also mandatory in patients with pain in the abdomen and epigastric region.

Overall, the language doesn't seem to be standard. I would recommend authors to undertake editing by an English native speaker.